Author’s response to reviews

Title: Chronic Persistent Horner’s Syndrome in Trigeminal Autonomic Cephalalgia Subtypes and Alleviation with Treatment: Case Reports

Authors:

Todd Rozen (Rozen.Todd@mayo.edu)

Matthew Kline (mtkline@gmail.com)

Version: 1 Date: 08 Jan 2019

Author’s response to reviews:

We appreciate the reviewer’s kind comments

The response to Reviewer 1’s comments are below

Reviewer 2 had no comments

There were no Editor comments

Author's response: In regard to the ptosis vs facial spasm or blepharospasm question -We agree the image (which has been compressed to not show the patient’s entire facial profile) almost looks like spasm based on the wrinkling around the left eye but at the bedside and by observation of ophthalmology there was no spasm involved -this was a ptosis alone without a miotic pupil thus we use the term a partial Horner’s syndrome.

To help clarify this issue we have included the following statement on page 4 of the manuscript and in the Figure legend “There was no evidence of blepharospasm or facial spasm noted on exam too mimic the ptosis”

Thank you